\newpage
\addcontentsline{toc}{chapter}{List of publications}
\bibliographypub{../../bibdb/bibdb}{}
\bibliographystylepub{thesispub}

% own publications
\nocitepub{zirkelbach11,zirkelbach10,zirkelbach09}
\nocitepub{zirkelbach2007,zirkelbach2006,zirkelbach2005}

%\renewcommand\bibname{List of publications}

